Author’s response to reviews

Title: Quantitative proteomic characterization of human sperm cryopreservation: using data-independent acquisition (DIA) mass spectrometry

Authors:

Longlong Fu (fullpcmu@163.com)
Qi An (276724023@qq.com)
Kaishu Zhang (1323961050@qq.com)
Ying Liu (1151294240@qq.com)
Yue Tong (fullcmu@126.com)
Jianfeng Xu (jianfengxu2016@163.com)
Fang Zhou (zhoufang16@126.com)
Xiaowei Wang (xwwang16@163.com)
Yin Guo (495424764@qq.com)
Wenhong Lu (wenhonglu16@163.com)
Xiaowei Liang (liangxw61@126.com)
Yiqun Gu (guyq16@163.com)

Version: 4 Date: 20 Nov 2019

Author’s response to reviews:

And written informed consent was obtained from all the sperm donors involved in the study.